Author's response to reviews

Title: Extramedullary hematopoiesis presenting as a compressive cord and cerebral lesion in a patient without a significant Hematologic Disorder

Authors:

Amir Saied Seddighi (invincible19152@gmail.com)
Afsoun Seddighi (invincible19152@yahoo.com)

Version: 7 Date: 31 March 2010

Author's response to reviews: see over
Dear Editor:

Thanks a lot for your guidelines to correct the format of our article. As your order, we corrected the grammatical writing errors and we changed the term muscle force to muscle strength. We added the requested information including the explanation that the patient did not smoke at all and the level of her erythropoietin was normal and arterial blood gas analysis showed no hypoxia and normal oxygen saturation. We have also omitted the dates of treatments and visits and added the conclusion section to the paper.

It would be a great honor if you accept this paper in your respectful journal.

Sincerely,

Dr. Amir Saied Seddighi
Dr. Afsoun Seddighi